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ABSTRACT 

Global energy requirements are increasing day by day at a significant rate, and this 

demand is being primarily met by fossil fuels like natural gas, coal, and fractions of petroleum. 

On one hand, there is a threat of depletion of the finite fossil energy resources; on the other 

hand, there is an equally imminent threat of global warming through non-judicious burning of 

fossil energy. Hydrogen has been suggested as an attractive and clean energy carrier depending 

on how it is produced, and serves as a remedy to the depletion of fossil fuels and the associated 

environmental concerns. Aside from the major applications of hydrogen in the chemical, foods 

and refining industries, it can be used in fuel cells to power vehicles, as well as for other 

stationary and non-stationary applications.  

“On-board” or “distributed” processing of fuel to produce hydrogen, on demand, seems 

like an obvious concept, but is often not viable because of the fact that fuel reforming is 

endothermic, and energy needs to be supplied to the reforming reactor in a way so as to offset 

the drop in reactor performance due to the “cooling” induced by the reaction. One way of 

offsetting this issue is to create a heat-integrated design. However, such designs have 

exothermic and endothermic fronts that move in time and space, and hence the problem is non-

trivial. Effective design of such reactors require knowledge of the time and space movements 

of such fronts. Thus, the final "success" of this technology will depend largely on modeling, 

understanding and control of temperature excursions from the steady state situation.  

In this thesis, such process concepts are evolved using mathematical modeling of the 

ethanol steam reforming reactor, to produce hydrogen in a compact reformer under dynamic 

operation.  

In the first part of the study, various spatio-temporal evolutions of temperature and 

conversions in the reaction zones were explored through models developed for recuperative 

heat exchange and process integration. Further, a design alternative was investigated for the 

effective transport of heat. The species flow in exothermic section was shifted in order to study 

its impact on temperature fronts. The extensive modeling work revealed the way in which the 

temperature fronts were expected to evolve, and how that affected the design alternatives. 

Alternate designs by conducting the reforming reaction in monolith microreactors, rather than 

packed beds, was also evolved. A design parameter ‘volumetric reactor activity’ to define an 

optimal extruded and washcoated monolith configuration was given. 
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To have a complete picture of the ethanol steam reforming at the reactor level, catalyst 

deactivation was also accounted. Several alternative inlet feed configuration schemes 

(distributed feed) and recycle scheme were also explored. 

 

 



 
 

सार 

वै श्विक ऊर्जा आिवयकतजओं को दिन-प्रततदिन एक महत्िपूर्ा िर से बढजयज र्ज रहज ह , 

और यह मजंग मुख्य रूप से प्रजकृततक ग स, कोयले और पेट्रोललयम के अशं र् से र्ीिजवम ईंधन 

से लमलती ह । एक ओर, पररलमत र्ीिजवम ऊर्जा संसजधनों की कमी कज खतरज ह ; िसूरी तरफ, 

र्ीिजवम ऊर्जा के ग र-न्यजयी र्लजने के मजध्यम से ग्लोबल िजलमिंग कज एक समजन आसन्न 

खतरे ह । हजइड्रोर्न को एक आकर्ाक और स्िच्छ ऊर्जा िजहक के रूप में सुझजयज गयज ह  कक 

यह क से त यजर ककयज र्जतज ह , और र्ीिजवम ईंधन की कमी और संबंधधत पयजािरर्ीय ध तंजओं 

के उपजय के रूप में कजया करतज ह । रसजयन, खजद्य पिजर्ा और शोधन उद्योगों में हजइड्रोर्न के 

प्रमुख अनुप्रयोगों के अलजिज, यह ईंधन कोलशकजओ ंमें बबर्ली के िजहनों के सजर्-सजर् अन्य 

श्स्र्र और ग र-श्स्र्र अनुप्रयोगों के ललए इस्तेमजल ककयज र्ज सकतज ह । 

मजंग पर, हजइड्रोर्न कज उत्पजिन करने के ललए न न बोड"ान यज नतितररतन प्रसंस्करर् की 

प्रकियज एक स्पष्ट अिधजरर्ज की तरह दिखती ह , लेककन इस तथ्य की िर्ह से अक्सर व्यिहजया 

नहीं होतज ह  कक ईंधन सधुजर करनज एंड"ोस्टमी ह , और सुधजर ररएक्टर को ऊर्जा की आपूतत ा की 

र्जनी  जदहए एक तरह से प्रततकियज से प्रेररत नशीतलनन के कजरर् ररएक्टर प्रिशान में धगरजिट 

को  फसेट करने के ललए। इस मुदे्द को  फसेट करने कज एक तरीकज एक गमी-एकीकृत 

डड"र्जइन बनजनज ह  हजलजंकक, इस तरह के डड"र्जइनों में एक्सओरे्रलमक और एडं"ोरे्लमक मो ों होत े

हैं र्ो समय और स्र्जन पर र्जते हैं, और इसललए समस्यज क्षुद्र ह । ऐसे ररएक्टरों के प्रभजिी 

डड"र्जइन को ऐसे मो ों के समय और अतंररक्ष आिंोलनों के बजरे में ज्ञजन की आिवयकतज होती 

ह । इस प्रकजर, इस तकनीक कज अतंतम नसफलतजन श्स्र्र श्स्र्तत की श्स्र्तत से मॉड"ललगं, समझ 

और तजपमजन भ्रमर् के तनयंत्रर् पर कजफी हि तक तनभार करेगज। 

इस कजम में, गततशील  परेशन के तहत एक कॉम्प क्ट सुधजरक में हजइड्रोर्न कज उत्पजिन 

करने के ललए, एरे्नॉल स्टीम ररफ क्टर सुधजर करने के गणर्तीय मॉड"ललगं कज उपयोग करके 

इस तरह की प्रकियज अिधजरर्जएं तिकलसत की र्जती हैं। 



 
 

 
 

अध्ययन के पहले भजग में, प्रततकियज क्षेत्रों में तजपमजन और रूपजंतरर् के तिलभन्न स्र्जतनक-

अस्र्जयी तिकजस ने शोधनशील गमी तितनमय और प्रकियज एकीकरर् के ललए तिकलसत मॉड"ल 

के मजध्यम से पतज लगजयज गयज। इसके अलजिज, गमी के प्रभजिी पररिहन के ललए एक डड"र्जइन 

तिकल्प की र्जं  की गई। एक्स्टोमजयलमक खडं" में प्रर्जततयों के प्रिजह को तजपमजन मो ों पर 

इसके प्रभजि कज अध्ययन करने के ललए स्र्जनजंतररत ककयज गयज र्ज। व्यजपक मॉड"ललगं कजया 

ने श्र्स तरह से तजपमजन के मो ों के तिकजस की उम्मीि की र्ी, और क से डड"जजइन तिकल्प 

को प्रभजतित ककयज प क बेड"ों की बर्जय, मोनोललर् मजइिोएरेक्टसा में सुधजर की प्रततकियज कज 

आयोर्न करके ि कश्ल्पक डड"र्जइन भी तिकलसत हुए रे्। एक इष्टतम तनकलज हुआ  और उबकज 

हुआ मोनोलिथ तिन्यजस को पररभजतर्त करने के ललए एक डड"र्जइन प रजमीटर 'वोिूमेलरिक ररएक्टर 

गतततिधध' दियज गयज र्ज। 

ररएक्टर स्तर पर सुधजर में इरे्नॉल स्टीम की एक पूरी तस्िीर रखने के ललए, उत्प्रेरक 

तनश्ष्िय करने कज भी र्िजब दियज गयज र्ज। कई ि कश्ल्पक इनलेट फीड" तिन्यजस योर्नजए ं

(तितरर् फीड") और रीसजयकल स्कीम कज भी पतज लगजयज गयज। 
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NOMENCLATURE 

Lb length of reactor [m] 

Dp catalyst particle diameter [m] 

D reactor diameter [m] 

F molar flow rate [mol s-1] 

Ac cross-sectional area [m2] 

n reactor order [-] 

c specie concentration [mol m-3] 

Ea activation energy [kJ mol-1] 

RG gas constant [kJ mol-1K-1] 

H  heat of reaction [kJ mol-1] 

Cp specific heat [kJ kg-1K-1] 

pĈ  molar heat capacity [kJ mol-1K-1] 

pC
~

 volumetric heat capacity [kJ m-3K-1] 

V  volume of reactor [m3] 

W weight of catalyst [kg] 

C dimensionless concentration [-] 

R radius [m] 

r reaction rate [mol m-3 s-1] 

krxtr catalyst activity on reactor volumetric basis [s-1] 

ν volumetric flow rate [m3 s-1] 

rapp apparent reaction rate [mol m-3 s-1] 

rcat,V reaction rate [mol m-3
cat s

-1] 

U, h heat transfer coefficient [W m-2K-1] 

u velocity [m s-1] 

X conversion [-] 

q heat flux [W m-3] 

u velocity vector [m s-1] 

p pressure [Pa] 

I identity matrix [-] 

Nj molar flux [mol m-2s-1] 

k thermal conductivity [W m-1K-1] 

dl  characteristic diffusion length [m] 

vk  intrinsic rate constant [(mol-1 m3)(n-1) s-1] 
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Deff effective diffusivity [m2 s-1] 

mr  effective molecular radius [m] 

pr  pore radius [m] 

t time [s] 

0

Cc  maximum surface coke concentration [kg m-2] 

Cc  surface coke concentration at any instant [kg m-2] 

KD  Knudsen diffusion at any instant [m2 s-1] 

DM Molecular diffusivity [m2 s-1] 

CC  dimensionless coke concentration [-] 

CR  reaction rate of coke [mol m-3 s-1] 

T temperature [K] 

n normal vector [-] 

r, z space coordinates [m] 

dh hydraulic diameter [m] 

tw wall thickness [m] 

tc washcoat thickness [m] 

Da Damkohler number [-] 

dt tube diameter [m] 

RR recycle ratio [-] 

P perimeter of reactor tube [m] 

J dimensionless source [-] 

NT total number of distributed feed locations [-] 

N distributed feed location (1, 2, …, NT) [-] 

H Hydrogen 

CD Carbon dioxide 

CO Carbon monoxide 

M Methane 

E Ethanol 

S Steam  

  

Greek letters 

ρcat,b bulk density of catalyst [kg m-3] 

  particle voidage [-] 
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S  solid fraction [-] 

b  bed voidage [-] 

int  internal effectiveness factor [-] 

  thiele modulus [-] 

  dimensionless time [-] 

m  Particle tortuosity [-] 

  dimensionless length [-] 

  dimensionless heat of reaction [-] 

  dynamic viscosity [Pa s] 

  dimensionless temperature [-] 

C  bulk density of coke [kg m-3] 

  dimensionless activation energy [-] 

  

Superscripts 

0 initial 

c combustion 

r reforming  

* dimensionless 

‘ backward 

  

Subscripts  

i reaction 

j component 

cat catalyst 

s solid 

g gas 

in inlet 

B backbone 

S steam 

ss side stream 
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